Accounting Discretion in Goodwill Impairments: UK Evidence

Naser M. AbuGhazaleh

College of Business Administration, Gulf University for Science and Technology (GUST), P.O. Box 7207 Hawally, 320 93, Kuwait e-mail: abughazaleh.n@gust.edu.kw

Osama M. Al-Hares

College of Business Administration, Gulf University for Science and Technology (GUST), P.O. Box 7207 Hawally, 320 93, Kuwait e-mail: alhares.o@gust.edu.kw

Clare Roberts

University of Aberdeen Business School, Old Aberdeen, AB24 3QY, U.K. e-mail: c.roberts@abdn.ac.uk

Abstract

This study examines managers' use of discretion in determining goodwill impairment losses following the mandatory adoption of IFRS 3 "Business Combinations," and whether this discretion reflects opportunistic reporting by managers or the provision of their private information. Although IFRS 3 was issued to improve the accounting treatment for goodwill and provide users with more useful and value-relevant information regarding the underlying economic value of goodwill, it has been criticized on the grounds of the managerial discretion inherent in impairment testing. Therefore, *ex-ante*, it is unclear how the impairment-only approach has affected the reporting of goodwill impairment losses. After controlling for economic factors, empirical results reveal that managers are exercising discretion in the reporting of goodwill impairments following the adoption of IFRS 3. Specifically, goodwill impairments are more likely to be associated with recent CEO changes, income smoothing and "big bath" reporting behaviors. However, the results also indicate that goodwill impairments are strongly associated with effective governance mechanisms suggesting that managers are more likely to be exercising their accounting discretion to convey their private information about the underlying performance of the firm rather than acting opportunistically. These inferences are robust to various modeling specifications and variable definitions, suggesting that IFRS 3 has provided managers with a framework to reliably convey their private information about future cash flows consistent with the IASB's objectives in developing the impairment standard.

1. Introduction

On March 31, 2004, the International Accounting Standards Board (IASB), seeking international convergence and global harmonization,

The authors are grateful for helpful comments received at the 2009 British Accounting Association Conference, University of Dundee.

^{© 2011} Blackwell Publishing Ltd., 9600 Garsington Road, Oxford OX4 2DQ, UK and 350 Main Street, Malden, MA 02148, USA.

followed the U.S. Financial Accounting Standards Board (FASB) and issued IFRS 3, *Business Combinations*, (IASB, 2004a). IFRS 3 eliminates the use of the pooling of interests method and prohibits the amortization of goodwill. Instead it requires the testing for impairment annually or more frequently if events or changes in circumstances indicate that the asset might be impaired. With the transition to international reporting standards, U.K. firms listed in the main market had to discontinue amortizing goodwill and account for it using IFRS 3 since 2005.

The impairment approach to goodwill was introduced with the intent of improving the information content of reported acquired goodwill and reducing the managerial flexibility afforded by the former trigger-based standards. However, this approach has been criticized by academics, practitioners, and dissenting IASB members because of the managerial discretion inherent in the process of testing goodwill for impairment, and the resulting blending of acquired goodwill and internally generated goodwill (Massoud and Raiborn, 2003; Watts, 2003). Standard setters suggest managers will use this discretion to convey their private information on future cash flows, resulting in impairments that are more reflective of the firm's underlying economics.¹ Alternatively, managers may opportunistically exploit this unverifiable accounting discretion to extract rents from other contracting parties, resulting in impairments that are less reflective of the firm's underlying economics, thus mitigating the purported benefits of the impairment approach. Therefore, *ex-ante*, it is unclear how the impairment-only approach has affected the reporting of goodwill impairment losses, including the related managerial flexibility exercised in determining them.

Motivated by the above debate, the primary objective of this study is to examine managers' use of discretion in determining goodwill impairment losses following the mandatory adoption of IFRS 3, and whether this discretion reflects opportunistic reporting by managers or the provision of their private information. Using a sample of 528 firmyear observations, drawn from the top 500 U.K. listed firms for 2005 and 2006, the study employs a pooled multivariate tobit regression and investigates the extent to which proxies for economic impairment, managerial discretion, and effective corporate governance mechanisms explain the magnitude of goodwill impairment losses.

The study hypothesises that goodwill impairment losses are conceptually a function of economic factors underlying the performance of © 2011 Blackwell Publishing Ltd.

the firm, reporting incentives of managers, and constraints imposed by effective corporate governance mechanisms. Thus, managers should record a goodwill impairment loss if they detect that the recoverable value of a cash-generating unit has declined below its carrying value. However, based on the managerial discretion inherent in IFRS 3 and consistent with the contracting theory of accounting (Watts and Zimmerman, 1986, 1990; Holthausen, 1990; Christie and Zimmerman, 1994), managers may overstate, understate, or simply not recognize an existing economic impairment depending on their reporting incentives. Although this behavior may be opportunistically driven, managers disciplined by effective governance mechanisms are less likely to act opportunistically but instead exercise their accounting discretion to convey their private information about the true value of the firm. Hence, goodwill impairments strongly associated with effective governance mechanisms are more likely to reflect the provision of managers' private information rather than their acting opportunistically.

After controlling for economic factors, empirical results reveal that managers are exercising discretion in the reporting of goodwill impairment losses following the adoption of IFRS 3. Specifically, goodwill impairments are more likely to be associated with recent CEO changes, income smoothing and "big bath" reporting behaviors. However, they also reveal that impairments are strongly associated with effective governance mechanisms measured by the percentage of independent directors on the board of directors, the number of board meeting, the percentage of shares owned by blockholders holding at least 10 per cent of outstanding shares, and the percentage of shares owned by executive and non-executive directors, suggesting that managers are more likely to be using the guidelines specified by IFRS 3 to convey their private information on future cash flows rather than acting opportunistically. Accordingly, the study provides evidence consistent with the IASB's objectives in developing the standard in that the results collectively suggest that IFRS 3 has provided firms with a framework to reliably reflect their underlying economic attributes. These inferences are robust to various modeling specifications and variable definitions. They are also robust to the principal component aggregation of the corporate governance variables.

This study contributes to the extant empirical research on asset and goodwill write-offs in two ways. First, it provides empirical evidence on the determinants of goodwill write-offs recorded on transition to IFRS 3 in the U.K. context which has not been examined before.

Second, it examines the effect of a wide group of governance indicators in providing incentives for managers to engage in high quality goodwill reporting practices. The use of multiple indicators to proxy for corporate governance can alleviate the measurement error associated with a single governance indicator and result in fewer correlated omitted variable problems which may yield more consistent regression coefficients (Larcker et al., 2007). However, without a clear theory about the complex, multi-dimensional nature of corporate governance or a conceptual basis for selecting the relevant governance variables, principal component analysis is also used in this study to factorize governance indicators into more reliable and valid factors that collectively capture different dimensions of corporate governance quality and yield more stable estimates.

The results of this study should be of interest to standard setters and policy makers as they suggest that managers are likely to utilize the discretion permitted in principles-based standards. They also highlight the importance of effective governance mechanisms in constraining managerial opportunism associated with such standards.

The remainder of this article is organized as follows. Section 2 discusses the background. Section 3 presents the main findings of prior research relating to write-offs. Section 4 develops the research hypotheses. Section 5 explains the research design employed to empirically test the hypotheses. Section 6 reviews descriptive statistics and empirical results. Section 7 presents additional tests and Section 8 concludes the study.

2. Accounting for Goodwill

Accounting for goodwill has long been a controversial issue in the United Kingdom, at least partly because of the quantities of goodwill that managers have to deal with (Nobes, 1992; Higson, 1998). The first attempt by the U.K. standard setter, SSAP 22 *Accounting for Goodwill*, in 1984 (ASC, 1984), required goodwill to be either written off against reserves or capitalized and amortized over an "appropriate" period. This attempt received enormous criticisms as it permitted two different accounting treatments that were "conceptually inconsistent" (Hussey and Ong, 2000). The debate in the late 1990s in the United Kingdom led to the establishment of FRS 10 *Goodwill and Intangible Assets* (ASB, 1997), whereby goodwill must be capitalized and amortized (with trigger-based impairment tests) under a rebuttable presumption

that its useful economic life does not exceed 20 years from the date of acquisition instead of the immediate write-off to reserves.²

Under IFRS 3, goodwill is no longer amortized but tested for impairment in accordance with IAS 36 (IASB, 2004b). According to IAS 36, goodwill acquired in a business combination should, from the acquisition date, be allocated to each of the acquirers' cash-generating units, or groups of cash-generating units that are expected to benefit from the synergies of the business combination. Each unit or group of units to which goodwill is allocated should represent the lowest level within the entity at which goodwill is monitored for internal management purposes; and it should not be larger than a segment based on either the entity's primary or secondary reporting format according to IAS 14 Segment Reporting (replaced by IFRS 8 Operating Segments on November 30, 2006).³ A cash-generating unit to which goodwill has been allocated shall be tested for impairment both annually and whenever there is an indication that the unit may be impaired. If the recoverable amount of the unit exceeds the carrying amount of the unit, the unit and the goodwill allocated to that unit are not impaired. If, instead, the carrying amount of the unit exceeds its recoverable amount, the entity must recognize an impairment loss.⁴ The recoverable amount of an asset or a cash-generating unit is the higher of its fair value less costs to sell and its value in use.

The impairment loss is allocated firstly to the goodwill of the cashgenerating unit (group of units) and then on a *pro rata* basis to the other assets within the unit (group of units), as long as it does not reduce any asset below the highest of its fair value less costs to sell, its value in use, and zero. The impairment loss is recognized immediately above the line in income from continued operations. Once recognized, IAS 36 prohibits the recognition of reversals of impairment losses for goodwill in subsequent periods.

Although IFRS 3 forces managers to perform annual goodwill impairment tests, it also provides the opportunity for accounting discretion by requiring managers to make a number of accounting choices. The most important of these are the determination of the cash-generating units, the subsequent allocation of goodwill to these units, and the recoverable amount estimates of the units. By exercising discretion inherent in IFRS 3, managers may, depending on their reporting incentives, overstate, understate, or simply not recognize an existing economic impairment loss by being selective with respect to the underlying choices they make when testing goodwill for impairment. This discretion may be used to convey managers' private information about future cash flows. Alternatively, it may be used opportunistically to extract rents from other contracting parties resulting in impairments that are less reflective of the firm's underlying economics.

3. Prior Research

The earlier asset write-off studies (Strong and Meyer, 1987; Elliott and Shaw, 1988; Zucca and Campbell, 1992) focus solely on managers' reporting incentives to take write-offs and fail to consider managers' incentives to take write-offs that reflect real declines in the value of assets. After controlling for economic factors, these studies report that write-offs are more likely to be associated with proxies for managerial discretion (recent management changes, "big bath" and income smoothing reporting behaviors). Francis et al. (1996) were among the first to examine whether write-offs are driven by managers' incentives to "manipulate" earnings or by changing economic circumstances of the firm. They find that both factors—asset impairment (poor past share performance) and managerial incentives (management changes)—are important determinants of asset write-offs. In a U.K.-based study, Kvaal (2005) finds that fixed asset impairment accounting under FRS 11, except that of goodwill, may be unbiased.

However, although these studies recognize that write-offs may be motivated by economic incentives, they "equate" managers' incentives with manipulation (opportunism) (Wilson, 1996, p. 176) and fail to consider the possibility that managers may be using their discretion efficiently to convey their private information on future cash flows. Rees et al. (1996) were among the first to explicitly examine the extent to which the discretion associated with write-offs may be value relevant rather than opportunistic. They find that pre-write-off earnings are, on average, significantly worse than industry medians, consistent with the "big bath" reporting behavior. However, they also find that write-off firms record significant negative abnormal accruals in the write-off year which do not reverse in subsequent years, suggesting that managers are responding to changes in economic circumstances rather than acting opportunistically. Riedl (2004) extends this approach and provides evidence that asset write-offs reported following the adoption of SFAS 121 in the United States are significantly associated with "big bath" reporting behavior. However, Riedl also finds that this behavior is more evident when the CEO is also the chairman of the board and suggests that this "big bath" behavior more likely reflects opportunistic reporting than managers providing their private information.

Studies examining the determinants of goodwill write-offs following the adoption of SFAS 142/Section 3062 in the United States and Canada were initiated by Beatty and Weber (2006) who examine managers' choices to report certain current goodwill write-offs belowthe-line or uncertain future write-offs in income from continuing operations. They find that proxies for managerial discretion (firms' equity market considerations, debt contracting, bonus, CEO turnover, and exchange delisting incentives) affect managers' decisions to accelerate or delay goodwill write-offs. In a similar study, Zang (2008) finds that debt contracting and management changes are important factors in determining the amount of SFAS 142 transitional goodwill impairment losses. In a Canadian context, where transitional goodwill write-offs are charged to retained earnings, Lapointe-Antunes et al. (2008) find that proxies for managerial discretion (CEO changes, managers' incentives to smooth ROA and ROE ratios, debt contracting, the value of managers' in-the-money exercisable stock options, the need for extra financing, and the cross-listing status of the firm) affect managers' incentives to overstate or understate transitional goodwill write-offs. They also report that financially literate and independent audit committee members effectively constrain managerial opportunism with respect to transitional goodwill write-offs. However, results of studies using United States and Canadian data from the transition period have to be interpreted with caution and may lack generalisability as, in recording write-offs, managers may be motivated by the special accounting treatments permitted in the transition period.⁵

In contrast to these studies, Godfrey and Koh (2009), Jarva (2009), and Ramanna and Watts (2009) examine goodwill impairments appearing as operating expenses in periods following the transition period. Godfrey and Koh (2009) find that goodwill impairments are negatively associated with firms' underlying investment opportunities and accounting returns and conclude that the introduction of SFAS 142 has enabled managers to provide information relevant to users of financial statements. Jarva (2009) finds that SFAS 142 goodwill impairments are associated with future expected cash flows as mandated by the standard. He also examines a sample of non-impairment firms in which there are indications that goodwill is impaired and fails to find convincing evidence that these firms are opportunistically avoiding impairments. Jarva concludes that goodwill impairments are, on average, more closely related to economic factors than opportunistic behavior. On the other hand, Ramanna and Watts (2009) provide evidence that non-impairment of goodwill is increasing in firm characteristics predicted to be associated with greater managerial discretion (number and size of reporting units and unverifiable net assets in reporting units). They fail to confirm that this discretion is being used to convey managers' private information. In an international context, Verriest and Gaeremynck (2009) examine goodwill impairments for a limited sample of European listed firms following the adoption of IFRS 3. They find that firms with stronger corporate governance mechanisms measured by the amount of independent members on the board are more likely to engage in goodwill impairment.

Another stream of literature examines the information content and value relevance of goodwill impairment losses (e.g., Chen et al., 2004; Zang, 2008; Lapointe-Antunes et al., 2009; Li et al., 2010). These studies find negative correlations between goodwill write-offs and stock returns and conclude that the impairment-only approach to goodwill has improved the quality of reported information on goodwill by allowing managers to reliably convey their private future-cash-flow information to markets. However, Ramanna (2008, p. 255) casts doubts on the "net benefit" conclusions of these studies and provides alternative explanations for the perceived negative stock market reaction.⁶ In addition, Ramanna argues that these studies are focused primarily on explaining recorded impairments (p. 255) and suggests that examining the determinants of goodwill impairments for samples that include both impairers and non-impairers may be a better approach to arriving at conclusions on the "net benefits" of the impairment standards.

This study differs from relevant prior research in that it focuses on managers' use of discretion with respect to IFRS 3 goodwill impairments reported in the U.K. context which has not been examined before. The absence of special transitional accounting treatments in the United Kingdom suggests that goodwill impairments are less likely to be affected by managerial incentives specific to the transition period, and hence increases the generalizability of the results. In addition, the study not only considers impairers and non-impairers, but attempts to test for managers' information-efficient incentives in explaining the accounting choices they make by controlling for the effect of a wide group of governance mechanisms hypothesized to play a role in mitigating managerial opportunism. It also improves upon the analyses of

prior relevant research by applying principal component analysis to group the individual corporate governance variables into more reliable and valid factors that collectively capture different dimensions of corporate governance quality and yield more stable estimates.

4. Hypotheses Development

4.1. Actual (Economic) Impairment Factors

Wilson (1996, p. 172) argues that the credibility of the research findings of asset write-off studies depends on the extent to which the experimental design controls for economic impairment factors. Goodwill impairment is a result of the deteriorating economic performance of the acquired business. Following the guidelines of IFRS No. 3, managers should record a goodwill impairment loss if they detect that the recoverable value of a cash-generating unit has declined below its carrying value. An ideal economic factor would include the managers' unbiased expectations regarding the future performance of cash-generating units that include goodwill (Riedl, 2004). However, as managers' expectations are not observable and as no financial information is publicly available at the cash-generating unit's level, the current study follows previous researchers (e.g., Francis et al., 1996; Riedl, 2004; Beatty and Weber, 2006; Lapointe-Antunes et al., 2008; Zang, 2008) and uses empirical proxies to capture the economic impairment of firm-wide goodwill. The actual impairment factors attempt to explain the real impairment of goodwill before discussing the attributes of managerial discretion. The above discussion results in the following hypothesis (stated in the alternative form):

H1: Ceteris paribus, there is a significant association between actual factors of impairment and reported goodwill impairment losses.

Six empirical proxies (book to market, size of goodwill, number of cash-generating units, change in turnover, change in operating cash flows, and ROA) are used, as discussed in Section 5.

4.2. Proxies for Managerial Discretion

Based on the managerial discretion inherent in IFRS 3 and consistent with implications of the contracting theory of accounting (Watts and Zimmerman, 1986, 1990), the study expects that managers are likely to utilize the discretion permitted by IFRS 3 in determining goodwill impairments. The following incentives are hypothesized to be associated with managers' discretionary behavior when they review goodwill for impairment.

4.2.1. Leverage. Watts and Zimmerman (1986, pp. 215–216) argue that managers of highly leveraged firms have incentives to select incomeincreasing accounting methods and estimates to avoid costly violations of debt covenants. In this scenario, it is expected that highly leveraged firms are less likely to record goodwill impairment losses to avoid costly violations of debt covenants (e.g., Riedl, 2004; Beatty and Weber, 2006; Lapointe-Antunes et al., 2008; Zang, 2008). An alternative argument regarding the role of debt argues that highly levered firms are likely to have the value of their assets under scrutiny from debt holders which may act as a disciplining device against opportunism and force the recognition of existing impairments that reflect the underlying performance of the firm (e.g., Strong and Meyer, 1987; Elliott and Shaw, 1988; Zucca and Campbell, 1992).

Based on the competing arguments and conflicting empirical results, the current study examines the relationship between reported goodwill impairment losses and leverage but does not predict the direction of the association. The above discussion results in the following sub hypothesis (stated in the alternative form):

H2a: Ceteris paribus, there is a significant association between the level of debt and reported goodwill impairment losses.

4.2.2. "Big bath" and income smoothing. Prior studies provide evidence that managers may use the discretion afforded by accounting standards to overstate impairment losses by taking "big bath" charges or by smoothing the reported earnings when they have incentives to do so (e.g., Zucca and Campbell, 1992; Francis et al., 1996; Rees et al., 1996; Riedl, 2004). Zucca and Campbell (1992, p. 36) suggest that "big bath" via asset write-offs is characterized by periods in which pre-write-off earnings are already below expected earnings. Managers may undertake a "big bath" in such periods to boost future earnings and provide a signal that "bad times" are behind them and better times will follow (Zucca and Campbell, 1992; Alciatore et al., 1998).

Income smoothing *via* asset write-offs is characterized by periods in which pre-write-off earnings are higher than expected. By recording © 2011 Blackwell Publishing Ltd.

write-offs during such periods, reported earnings will be closer (but not less than) the level expected (Zucca and Campbell, 1992, p. 36). Finally, Kirschenheiter and Melumad (2002) present a single model in which both "big bath" and income smoothing phenomena are part of an equilibrium reporting strategy. They argue that managers wish to report higher earnings in order to convey higher long-run earnings. However, if the news is bad, managers have incentives to under-report earnings by the maximum amount possible, preferring to take a "big bath" in the current period in order to reduce the inferred precision of the earnings surprises by smoothing earnings in order to raise the inferred precision of the earnings number. In both cases, this reporting behavior maximizes the value of the firm.

Based on the above arguments and consistent with prior research, the current study expects that managers may exercise their accounting discretion to *overstate* goodwill impairment losses by taking "big bath" charges (smoothing reported earnings) when pre-write-off earnings are abnormally low (high). The above discussion results in the following sub hypothesis (stated in the alternative form):

H2b: Ceteris paribus, firms with abnormally low and abnormally high pre-write-off earnings are more likely to report higher amounts of goodwill impairment losses.

4.2.3. Management change. Prior studies provide evidence that incoming CEOs have incentives to "take a bath" in the year of the executive change (e.g., Strong and Meyer, 1987; Elliott and Shaw, 1988; Francis et al., 1996; Riedl, 2004; Kvaal, 2005; Beatty and Weber, 2006; Lapointe-Antunes et al., 2008; Masters-Stout et al., 2008; Zang, 2008) as the reported low earnings may be blamed on the old management, and the historical bases for future comparison will be reduced. In addition, relieving future income of these charges enhances the opportunity for showing improved earnings and firm performance in later periods (Moore, 1973).

An alternative argument suggests that the positive association between impairments and changes in management may reflect true economic changes as opposed to managerial opportunism, as new management may exercise greater scrutiny over existing assets or change the firm's strategic focus, resulting in impairments (Francis et al., 1996; Wilson, 1996; Riedl, 2004). A final argument consistent with new managers of poorly performing firms taking impairments that reflect the true underlying economics of the firm suggests that management may be changed as a consequence of poor performance that necessitates goodwill impairment losses (Murphy and Zimmerman, 1993; Fields et al., 2001). To the extent the selected economic factors used in the regression equation already control for the underlying economic performance of the firm, this variable may capture new CEOs' reporting incentives to "take a bath" when they are appointed.

Based on the above arguments and consistent with the results of prior research, the current study expects that a greater amount of goodwill impairment losses will be reported by firms that experience a recent change in the CEO. The previous discussion results in the following sub hypothesis (stated in the alternative form):

H2c: Ceteris paribus, firms that experience a recent change in CEO are more likely to report higher amounts of goodwill impairment losses.

4.3. Corporate Governance

Wilson (1996, p. 176) argues that most of the proxies for managers' reporting incentives are also good proxies for economic impairment, suggesting that researchers should not equate managers' use of discretion with opportunism and should explicitly attempt to control for managers' information-efficient incentives in explaining the accounting choices they make. Consequently, this study controls for the effect of strong governance mechanisms hypothesized to constrain opportunism and restrict managers' ability to report write-offs that differ from predicted economic losses. Managers disciplined by effective governance mechanisms are less likely to act opportunistically but instead use their accounting discretion to convey their private information. Alternatively, given an incentive to manipulate, having weak governance structures is more likely to allow managers to opportunistically exploit their accounting discretion resulting in impairments that are less reflective of the firm's underlying economics.

Prior studies suggest that effective governance mechanisms help improve the quality of financial reporting. In particular, they provide evidence that firms with stronger governance mechanisms are less

likely to engage in earnings management, commit fraud or be subject to SEC enforcement actions for allegedly manipulating earnings (Warfield et al., 1995; Beasley, 1996; Dechow et al., 1996; Bushee, 1998; Chtourou et al., 2001; Klein, 2002; Koh, 2003; Xie et al., 2003; Peasnell et al., 2005; Mulgrew and Forker, 2006; Ebrahim, 2007). They also provide evidence that effective governance practices are positively related with disclosure quality (e.g., Karamanou and Vafeas, 2005). Furthermore, Verriest and Gaeremynck (2009) report that firms with stronger corporate governance mechanisms measured by the amount of independent members on the board are more likely to engage in goodwill impairment. Hence, motivated by the above and following prior research, this study expects goodwill write-offs to have significant positive associations with effective governance mechanisms under the information-efficient perspectives of accounting choice.⁷ Shareholders will not expect boards to constrain accounting choices credibly used by managers to signal their private information about future cash flows (Healy and Wahlen, 1999). This leads to the following hypothesis (stated in the alternative form):

H3: Ceteris paribus, firms with stronger governance mechanisms are more likely to report higher amounts of non-opportunistic good-will impairment losses.

Six empirical proxies (board independence, separate chairman, activity, blockholders, and shares held by executive and non-executive directors) are used, as discussed in Section 5.

5. Research Design

A one-stage test⁸ using a multivariate pooled tobit regression⁹ is used in this study to examine the determinants of goodwill impairment losses in United Kingdom. The model controls for the time period by adding a year-end dichotomous variable (*YEND*) that takes the value of 1 for the 2005 year-end firm observations, and 0 for the 2006 yearend firm observations. Data on goodwill impairments and corporate governance variables are hand-collected from the annual reports of the sample firms, while firm-specific financial variables are collected from the *Hemscott Premium Database*, supplemented by the firms' annual reports when necessary. Finally, financial statements prepared in a currency different from pounds sterling are translated into pounds using the exchange rate at the balance sheet date. The following model is used to implement the analysis:

$$\begin{split} GIL &= \alpha + \beta_1 B/M + \beta_2 GWA + \beta_3 CGU + \beta_4 \Delta TURNOVER + \beta_5 \Delta OCF \\ &+ \beta_6 ROA + \beta_7 DEBTRATIO + \beta_8 BATH + \beta_9 SMOOTH \\ &+ \beta_{10} \Delta CEO + \beta_{11} BINDEP + \beta_{12} SEPCHAIR + \beta_{13} BACTIVITY \\ &+ \beta_{14} BLOCK + \beta_{15} EXEOWN + \beta_{16} NONEXEOWN + \beta_{17} ADD \\ &+ \beta_{18} USCLIST + \beta_{19} YEND + \beta_{20} SIZE + e \end{split}$$

where:

GIL	=	firm <i>i</i> 's reported goodwill impairment loss (expressed as a positive number) deflated by total assets at the end of $t = 1$:
B/M	=	firm \vec{r} 's book value of equity (adjusted for goodwill write- offs) divided by market value of equity at the end of t :
GWA	=	firm <i>i</i> 's opening carrying value of goodwill deflated by total assets at the end of $t - 1$:
CGU	=	a dichotomous variable equal to 1 if firm <i>i</i> has more than one cash-generating unit at the end of <i>t</i> , and 0 otherwise;
$\Delta TURNOVER$	=	the change in turnover for firm <i>i</i> from period $t - 1$ to t deflated by total assets at the end of $t - 1$;
$\triangle OCF$	=	the change in operating cash flows for firm <i>i</i> from period $t - 1$ to <i>t</i> deflated by total assets at the end of $t - 1$;
ROA	=	the return on assets for firm <i>i</i> at the end of $t - 1$ (measured as pre-tax profit divided by total assets);
DEBTRATIO	=	firm \vec{i} 's total debt at the end of $t - 1$ divided by total assets at the end of $t - 1$;
BATH	=	the change in firm t 's pre-write-off earnings from $t - 1$ to t deflated by total assets at the end of $t - 1$, when this change is below the median of non-zero negative values of this variable, and 0 otherwise:
SMOOTH	=	the change in firm <i>i</i> 's pre-write-off earnings from $t - 1$ to t deflated by total assets at the end of $t - 1$, when this change is above the median of non-zero positive values of this variable, and 0 otherwise:
ΔCEO	=	a dichotomous variable equal to 1 if firm <i>i</i> experiences a change in CEO in $t = 1$ or t, and 0 otherwise:
BINDEP	=	the number of independent non-executive directors divided by the total number of board members:
SEPCHAIR	=	a dichotomous variable equal to 1 if the roles of chairman and CEO are separate, and 0 otherwise:
BACTIVITY	=	the number of meetings of the board of directors during the financial year;
BLOCK	=	the cumulative percentage of outstanding common shares held by blockholders holding at least 10 per cent of outstanding shares and who are not part of the board of directors:
EXEOWN	=	the total number of beneficial and non-beneficial common shares held by executive directors divided by the total number of outstanding common shares;

NONEXEOWN	=	the total number of beneficial and non-beneficial common shares held by non-executive directors divided by the total number of outstanding common shares:
ADD	=	a dichotomous variable equal to 1 if firm i has additions to its goodwill due to acquisitions during the financial year, and 0 otherwise;
USCLIST	=	a dichotomous variable equal to 1 if firm <i>i</i> is cross-listed on the U.S. <i>NYSE</i> or <i>NASDAO</i> , and 0 otherwise;
YEND	=	a dichotomous variable equal to 1 for the 2005 year-end firm observations, and 0 otherwise; and
SIZE	=	the natural logarithm of total assets at the end of $t - 1$.

Similar to Francis et al. (1996), Riedl (2004), Lapointe-Antunes et al. (2008), and Zang (2008), the dependent variable *GIL* is firm *i*'s reported goodwill impairment loss (expressed as a positive number) deflated by total assets at the end of t - 1.¹⁰

To test H1, six variables are used in the model to proxy for the economic impairment of goodwill. These variables are measured at the *firm level* and attempt to capture the actual impairment of *firm-wide* goodwill due to the unavailability of financial data to measure the actual impairment at the cash-generating unit level.¹¹

The first three variables are intended to proxy for the characteristics of goodwill (B/M, GWA, CGU). The first proxy, B/M, treats the whole firm as one cash-generating unit. Firms with a higher book-to-market ratio are expected to report more goodwill impairment losses. Following Zang (2008) and Lapointe-Antunes et al. (2008), the second proxy, GWA, is measured as the opening carrying value of goodwill deflated by total assets at the end of t - 1.¹² A firm with a greater amount of goodwill in its asset composition may report more goodwill impairment losses because the relative amount of goodwill exposed to impairment tests is greater (Zang, 2008). Therefore, the study expects a positive association between goodwill impairment losses and GWA. The third variable, cash-generating units, CGU,¹³ may have an impact on the likelihood of an impairment loss being recognized (Beatty and Weber, 2006). On one hand, firms with more than one cash-generating unit are expected to carry out more impairment tests and thus may report higher goodwill impairment losses because an existing loss in one unit cannot be netted against an increase in another unit (Schneider, 2001). Managers of firms with multiple cash-generating units may also have more flexibility to use their write-off discretion to overstate goodwill impairments (take a bath or smooth reported income) by allocating the greater part of goodwill to cash-generating units that are

expected to decrease in value. Alternatively, this discretion may be used to understate or avoid goodwill impairments by allocating the greater part of goodwill to cash-generating units that are expected to increase in value and hence lower the probability of recognizing good-will impairment losses. As a result, the current study does not predict a sign on *CGU*.

Similar to Francis et al. (1996), Riedl (2004), Lapointe-Antunes et al. (2008), and Zang (2008), the next three economic variables (\(\Delta TURNOVER, \Delta OCF, ROA) control for firm-specific past performance and firm-specific change in performance. The first variable, $\Delta TURNOVER$, captures accrual-related performance attributes (Riedl, 2004). In addition, it represents a gross measure of firm performance, which reflects more of the recoverability of goodwill's value. The second variable, $\triangle OCF$, captures cash-related performance attributes (Riedl, 2004). In addition, it represents a net measure of performance, which reflects more of the return on investment in goodwill (Riedl, 2004). Cash flows are expected to be a key economic driver that determines the amount of any goodwill impairment loss since value-in-use estimates are highly dependent on cash flow projections. Generating lower cash flows than expected increases the likelihood that impairment charges will be required. The third variable, ROA, captures the firm's prior profitability. It is expected that the poorer the firm's past performance, the greater the magnitude of reported goodwill impairment losses (Francis et al., 1996; Zang, 2008). As a result, the study predicts a negative sign on $\Delta TURNOVER$, ΔOCF , and ROA.

To test *H2a*, *H2b*, and *H2c*, four variables are used to proxy for incentives managers may face in recording goodwill write-offs (*DEBT-RATIO*, *BATH*, *SMOOTH*, $\triangle CEO$).

The first variable, *DEBTRATIO*, is measured as the firm's total debt at the end of t - 1 divided by total assets at the end of t - 1. Total debt is collected from the *Hemscott Premium Database* and defined as the sum of short-term (debt in current liabilities) and long-term debt. Unlike Riedl (2004) and Beatty and Weber (2006), this study does not use the details of firms' actual debt contracts and covenants to differentiate between private and public debt and uses a crude proxy (debt-to-asset ratio) to proxy for the tightness and proximity of firms to violation of their debt covenants. According to *H2a*, the current study does not predict a sign on *DEBTRATI*.

Similar to Riedl (2004), the second variable, *BATH*, is measured as the change in firms' pre-write-off earnings from t - 1 to t deflated by © 2011 Blackwell Publishing Ltd.

total assets at the end of t - 1, when this change is below the median of non-zero negative values of this variable, and 0 otherwise. The third variable, *SMOOTH*, is measured as the change in firms' pre-write-off earnings from t - 1 to t deflated by total assets at the end of t - 1, when this change is above the median of non-zero positive values of this variable, and 0 otherwise. According to H2b, the current study expects a negative (positive) association between reported goodwill impairment losses and *BATH* (*SMOOTH*).

Similar to previous studies (e.g., Lapointe-Antunes et al., 2008), the fourth variable, ΔCEO , is a dichotomous variable equal to 1, if the firm experiences a change in CEO in t - 1 or t, and 0 otherwise.¹⁴ The ΔCEO variable is hand-collected from the sample firms' annual reports. According to H2c, the current study expects a positive association between reported goodwill impairment losses and ΔCEO .

To test H3, six variables are used to proxy for effective corporate governance mechanisms (*BINDEP*, *SEPCHAIR*, *BACTIVITY*, *BLOCK*, *EXEOWN*, and *NONEXEOWN*). Consistent with H3, a positive sign is predicted on all six variables.

The first variable, *BINDEP*, is measured as the number of independent non-executive directors divided by the total number of board members. Directors' independence is measured according to firms' disclosures in their annual reports. Prior studies provide evidence that firms with a higher proportion of independent directors on their boards are less likely to engage in accruals earnings management, commit fraud or be subject to SEC enforcement actions for allegedly manipulating earnings (e.g., Beasley, 1996; Dechow et al., 1996; Xie et al., 2003; Peasnell et al., 2005).

The second variable, *SEPCHAIR*, is a dichotomous variable equal to 1 if the roles of chairman and CEO are separate and 0 otherwise. Fama and Jensen (1983) recommend that the roles of the chairman and CEO be separated, in order to limit the power of the CEO to expropriate the shareholders' interests. Jensen (1993, p. 866) also argues that without the direction of an independent chairman, it is much more difficult for the board to perform its critical monitoring function.

The third variable, *BACTIVITY*, as measured by the frequency of board meetings, is considered an important dimension of board monitoring operations (Vafeas, 1999; Xie et al., 2003), and the U.K. Combined Code (FRC, 2003, p. 5) recommends that the board should meet sufficiently regularly to discharge its monitoring duties effectively.

The fourth variable, BLOCK, is measured as the cumulative percentage of outstanding common shares held by blockholders holding at least 10 per cent of outstanding shares and who are not part of the board of directors¹⁵ (Beasley, 1996; Bushee, 1998; Chtourou et al., 2001; Lapointe-Antunes et al., 2008). Large shareholders have been viewed by prior studies as an alternative external governance mechanism which actively or passively monitors management's actions. Shleifer and Vishny (1986, 1997) suggest that large shareholders, in view of their significant equity stake in the firm, have greater incentives to collect information and monitor management's actions thereby avoiding the traditional free rider problem. Prior research also suggests that institutional shareholders in the United Kingdom are likely to be more active in their monitoring capacity than their U.S. counterparts as they face fewer restrictions with respect to their ownership and monitoring actions (Shleifer and Vishny, 1997; Short and Keasey, 1999). Short and Keasey (1999, p. 83) argue that much of the monitoring of companies by U.K. institutions takes place in a private "behind the scenes" fashion, which allows institutions to take joint actions to constrain managerial opportunism without drawing public attention to the fact. They further argue that the geographical clustering of the institutional shareholders within the City of London leads to an effective but informal monitoring of the U.K. listed companies.

The fifth variable, EXEOWN, is measured as the total number of common shares held by executive directors divided by the total number of outstanding common shares. The belief that managerial ownership helps to reduce agency costs and restrict managerial opportunism in agency settings arising from the separation of corporate ownership and control is well documented in the agency theory (Jensen and Meckling, 1976; Jensen, 1993). Alternatively, in a U.S.-based study, Morck et al. (1988) suggest that high levels of managerial ownership may provide managers with a deeper "entrenchment" because, through greater ownership levels, managers are able to exert greater control over decision making without fear of discipline from other external shareholders. However, in a U.K.-based study, Short and Keasey (1999) suggest that the greater institutional monitoring and the lesser ability to mount takeover defenses within the United Kingdom leads to management becoming entrenched at higher levels of ownership in the United Kingdom compared to their U.S. counterparts. The empirical results of Short and Keasey (1999) provide strong evidence of a consistently significant and positive association between performance and managerial ownership at

high levels of ownership. This is consistent with the hypothesis that the "convergence of interest" effect dominates the "entrenchment effect" at high levels of ownership in the United Kingdom.

The sixth variable, *NONEXEOWN*, is measured as the total number of common shares held by non-executive directors divided by the total number of outstanding common shares. Share ownership is expected to provide independent non-executive directors with greater incentives to monitor executive directors than other independent directors with insignificant shareholdings (Jensen, 1989, 1993).

Finally, four control variables are included in the model. ADD is a dichotomous measure of additions to goodwill due to acquisitions during the financial year. ADD may proxy for merger activity as firms active in mergers and acquisitions may be more inclined to impair goodwill associated with prior unprofitable acquisitions as the decrease in the carrying value of goodwill will be offset by the additions to goodwill during the year. Furthermore, the requirements to review goodwill arising on acquisition for impairment as soon as possible after the acquisition date may encourage managers to immediately identify and write off any overpayments relating to acquisitions that take place during the financial year. USCLIST is a dichotomous measure of U.S. cross-listing. Cross-listed firms may be more familiar with goodwill impairment tests because they have been required to carry out such tests under the requirements of SFAS 142 since 2001. Such firms are also likely to face a stricter enforcement regime resulting in more conservative U.K. earnings (Huijgen and Lubberink, 2005). Therefore, cross-listed firms may be more likely to report IFRS 3 goodwill impairments than other firms.¹⁶ As a result, the current study predicts a positive association between reported goodwill impairments and both ADD and USCLIST. YEND controls for the time period. Finally, SIZE is measured as the natural logarithm of total assets at the end of t - 1, Following prior studies (e.g., Beatty and Weber, 2006; Zang, 2008), the current study does not predict a sign on SIZE.

6. Empirical Results

6.1. Sample Construction, Year and Industry Composition

Table 1 presents the sample construction process. Firstly, the top 500 U.K. firms by total market capitalization as listed by the *Financial Times* at March 30, 2007 are selected for the 2005 and 2006 financial

	Firm-year observations
Top 500 U.Klisted firms by market capitalization (as listed by the <i>Financial Times</i> at March 30, 2007) for the 2005 and 2006 financial years	1,000
Observations belonging to the <i>Financials</i> industry	(254)
Observations listed on the Alternative Investment Market	(80)
Observations with no positive goodwill balances	(87)
Observations with insufficient/missing data	(51)
Final Sample	528
Goodwill impairers	109 (20.6%)
Non goodwill impairers	419 (79.4%)
Observations belonging to the 2005 year	256
Observations belonging to the 2006 year	272

Table 1. Sample Construction

Notes: This table presents the construction process for the final sample used to examine the determinants of IFRS 3 goodwill impairment losses.

years.¹⁷ This results in 1000 firm-year observations. Following prior research (e.g., Francis et al., 1996; Riedl, 2004), 254 observations belonging to the Financials industry are excluded since their financial reporting processes, as regulated industries, tend not to conform with other industries. Eighty observations listed on the Alternative Investment Market (AIM) are excluded as they were required to adopt the IFRS for the first time in 2007. Finally, 87 observations with no positive goodwill balances and 51 observations that do not have necessary data to run the tests are excluded. These procedures result in a final sample that consists of unbalanced data of 528 firm-year observations comprised of 109 write-off (20.6 per cent of sample) and 419 nonwrite-off observations (79.4 per cent of sample). The sample firms with a mean (median) market capitalization of £2338 (512) million (as listed by the Financial Times at March 30, 2007) are significantly larger than non-sample firms with mean (median) market capitalization of £1459 (327) million. Mean and median differences are significant at the 10 and 1 per cent level, respectively.¹⁸ The percentage of sample observations recording a goodwill impairment loss is 20.6 per cent.

The final sample consists of 256 observations belonging to the 2005 financial year comprised of 60 write-off (23.4 per cent) and 196 nonwrite-off observations; and 272 observations belonging to the 2006 financial year comprised of 49 write-off (18 per cent) and 223 non-write-off observations. The goodwill write-off percentage is not

significantly different between the 2 years included in the final sample of the study.

6.2. Descriptive Statistics

Table 2 provides descriptive statistics for the continuous variables used in the multivariate tobit regression examining the determinants of goodwill impairments losses, as well as the results of two-tailed *t*-tests of differences in means and two-tailed Mann-Whitney U-tests of differences in median. The mean (median) goodwill impairment charge for the final sample (N = 528) is £50.219 (£0) million. The mean (median) goodwill impairment charge for the write-off observations (N = 109) is £243.261 (£6.0) million. This mean goodwill impairment charge represents 1.8 per cent of the book value of total assets at the beginning of the financial year, and the median represents 0.30 per cent. Consistent with H1, write-off firms exhibit poorer financial performance than do non-write-off firms, reflected in a significantly lower mean and median for ROA and significantly lower medians for $\Delta TURNOVER$ and $\triangle OCF$. Consistent with the expectations of H3, they also have stronger governance mechanisms reflected in significantly higher means and medians for BINDEP, BACTIVITY, and BLOCK. However, contrary to the expectations of this study, executive directors hold significantly fewer shares in write-off firms as reflected by the significantly lower median for EXEOWN, supporting the "entrenchment hypothesis" that high levels of managerial ownership may be associated with managers acting opportunistically to avoid reporting goodwill impairment losses; and hence no support is provided for H3 with regard to EXEOWN on a univariate basis. In terms of control variables, consistent with prior research (Beatty and Weber, 2006; Zang, 2008), write-off firms are larger than non-write-off firms as reflected by the significantly higher mean and median for SIZE.

Table 3 provides descriptive statistics for the dichotomous variables used in the multivariate tobit regression. Consistent with prior research, write-off firms have more cash-generating units and experience more CEO changes than non-write-off firms, as reflected by the statistically significant differences on CGU and ΔCEO . In addition, write-off firms have more additions to their goodwill and have more observations cross-listed in the United States, as reflected by statistically significant differences on ADD and USCLIST.

	•										
		$All \ samp_{1}$ $(n = 528)$	le)	Ĩ	Write-ofj observation (n = 109)	f ns	N_i observa	on-write-o <u>j</u> ttions (n =	ff : 419)	Test of d (write-of non-wri	ifferences fs versus te-offs)
Variable	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD	Mean p-value	Median p-value
GILA	50.219	0.000	1,023.930	243.261	6.000	2,251.347	0.000	0.000	0.000		
GIL	0.004	0.000	0.020	0.018	0.003	0.040	0.000	0.000	0.000	<.000	<.000
B/M	0.331	0.302	0.258	0.362	0.329	0.334	0.323	0.298	0.234	.254	.436
GWA	0.190	0.137	0.177	0.197	0.150	0.173	0.188	0.134	0.178	.646	.413
<i>ATURNOVER</i>	0.113	0.090	0.461	0.075	0.046	0.284	0.123	0.106	0.497	.328	<.000 >
ΔOCF	0.020	0.011	0.098	0.007	0.004	0.130	0.023	0.013	0.087	.133	.036
ROA	0.084	0.075	0.120	0.060	0.056	0.117	0.090	0.083	0.120	.019	<.000 >
DEBTRATIO	0.249	0.224	0.191	0.245	0.213	0.159	0.249	0.227	0.198	.834	.691
BATH	-0.016	0.000	0.120	-0.017	0.000	0.100	-0.015	0.000	0.124	.887	.926
SMOOTH	0.029	0.000	0.057	0.033	0.000	0.071	0.028	0.000	0.052	.421	.919
BINDEP	0.498	0.5000	0.121	0.527	0.556	0.137	0.491	0.500	0.116	.005	<.000 >
BACTIVITY	9.057	9.000	3.036	10.064	9.000	4.126	8.795	8.000	2.624	.003	.002
BLOCK	12.034	10.390	15.015	14.905	11.040	18.300	11.287	10.220	13.963	.056	.087

Table 2. Descriptive Statistics - Continuous Variables

				nn T	10 2. (UUUU	inucu)					
		All sample $(n = 528)$			$ \begin{array}{l} Write-off\\ observations\\ (n = 109) \end{array} \end{array} $		N_i observa	on-write-oj ttions (n =	f 419)	Test of di (write-of) non-wri	fferences ^f s versus te-offs)
Variable	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD	Mean p-value	Median p-value
EXEOWN NONEXEOWN SIZE	0.038 0.014 6.889	0.002 0.001 6.700	$\begin{array}{c} 0.102 \\ 0.056 \\ 1.577 \end{array}$	0.034 0.017 7.316	0.001 0.0003 7.028	$\begin{array}{c} 0.111\\ 0.056\\ 1.698\end{array}$	$\begin{array}{c} 0.039 \\ 0.014 \\ 6.778 \end{array}$	0.002 0.001 6.600	$\begin{array}{c} 0.099 \\ 0.056 \\ 1.526 \end{array}$.661 .547 .001	.012 .150 .006
<i>Notes:</i> This table goodwill impairmé in medians. In this Variable definition (expressed as a po divided by market $\Delta TURNOVER$: thi cash flows for firm (measured as pre-t <i>BATH</i> : the change ian of non-zero ne total assets at the number of indeper of directors during per cent of outstal shares held by exee non-beneficial com logarithm of total	provides d ints losses, is table, the sition losses, GILA: the sitive num visitive num visitive num i from p a n i from p i from p gative valu end of t inter non- dent non- sithe finam nding shar cutive direc assets at th	escriptive statis as well as the bold values im the reported go ther) deflated b equity at the e riturover for inturover for invice $t - 1$ to fivide $t - 1$ to livide $t - 1$ to livide t varitie-off as pre-write-off a tes of this varitie - 1, when this rescutive direc cial year; BLO ces and are not escors divided t set held by non he end of $t - 1$	titics for the results of tv dicate significate significate significate signification odwill impa over the firm i from 1 to deflated be assets). DE analysis for assets), DE analysis for the sum of the signification of the signification of the signification of the signification of the overlap of the signification of th	continuous vo-tailed t_i cance. irment loss irment	variables used -tests of differe s amount (in n nd of $t - 1$; B s opening carr 1 to t deflated ets at the end ets at the end the stotal T in T is total T in T is total T in T is total T in T is total and number of ton- recentage of ou directors; EX outstanding c	d in the multiplication of the interval of the interval of I/M . firm i / M . firm i / M . for a large of i - 1; -1; -1; -1; -1; -1; -1; -1; -1; -1	Iltivariate 1 ans and tv pounds); <i>G</i> s book val s of goodwis soets at the <i>ROA</i> : the 1 ROA: the <i>i</i> + i at the end at the end firm <i>i</i> 's pre- to values o bers; <i>BAC</i> common sl e total nur ares; <i>NON</i>	cobit regressi vo-tailed Maa IL: firm $i's$ 1 ue of equity il deflated by the of $t -$ eturn on ass eturn on ass eturn on ass the of $t - 1$, w - t' the of $t - 1$, w - t' the arres held by inber of bene fuber of bene more of bene the of bene the of bene more of bene the of	on examin mn–Whitne eported gc (adjusted gc (adjusted sase) : ΔCF : total asse then this ch mumber o number o blockhold ficial and the total nu he total nu	ing the deter by <i>U</i> -tests of for goodwill impa for goodwill impa for goodwill test at the en- the change in a the change in the change is the en- the en-the en- the en-the en- the en-the en- the en-the en-the en- the en-the en-	minants of differences imment loss write-offs) i of $t - 1$; n operating d of $t - 1$; n operating d of $t - 1$; w the med-t, w the med-t, the board at least 10 ul common neficial and the natural

Variable	All sample (n = 528) proportion (%)	Write-off observations (n = 109) proportion (%)	Non-write-off observations (n = 419) proportion (%)	Chi-square test of difference (write-offs versus non-write-offs)
CGU	80.7	90.8	78	0.003
ΔCEO	25	35.8	22.2	0.004
SEPCHAIR	94.9	96.3	94.5	0.442
ADD	73.1	82.6	70.6	0.012
USCLIST	8.5	12.8	7.4	0.070
YEND	48.5	55	46.8	0.124

 Table 3. Descriptive Statistics – Dichotomous Variables

Notes: This table provides descriptive statistics for the dichotomous variables used in the multivariate tobit regression examining the determinants of goodwill impairments losses, as well as the results of two-tailed Chi-Square tests of differences in proportions. In this table, the bold values indicate significance.

Variable definitions: *CGU*: a dichotomous variable equal to 1 if firm *i* has more than one cash generating unit at the end of *t*, and 0 otherwise; ΔCEO : a dichotomous variable equal to 1 if firm *i* experiences a change in CEO in t - 1 or *t*, and 0 otherwise; *SEPCHAIR*: a dichotomous variable equal to 1 if the roles of chairman and CEO are separate, and 0 otherwise; *ADD*: a dichotomous variable equal to 1 if firm *i* has additions to its goodwill due to acquisitions during the financial year, and 0 otherwise; *USCLIST*: a dichotomous variable equal to 1 if firm *i* is cross-listed on the U.S. *NYSE* or *NASDAQ*, and 0 otherwise; and *YEND*: a dichotomous variable equal to 1 for the 2005 year-end firm observations, and 0 otherwise.

Table 4 presents the Pearson correlations for the variables used in the multivariate tobit regression. The table reveals that the proxies for the hypotheses are not highly correlated with one another or with the control variables. The highest pair-wise correlation coefficient is 0.510, suggesting that multicollinearity does not appear to be a problem in this study.¹⁹

Table 4 also reveals that the majority of actual (economic) impairment variables (B/M, GWA, ΔOCF , and ROA) are significantly correlated with IFRS 3 impairment charges (GIL) in the predicted direction. In terms of managers' reporting incentives, DEBTRATIO has a significant negative correlation with GIL. SMOOTH is significantly correlated with GIL in the predicted direction; and BATH and ΔCEO are correlated with GIL in the predicted direction; however, these correlations are not significant. In terms of corporate governance variables, BINDEP, BACTIVITY, and BLOCK are significantly correlated with GIL in the predicted direction; however, these correlated with GIL in the predicted direction. SEPCHAIR and NONEXE-OWN are correlated with GIL in the predicted direction; however, these correlations are not significant. Finally, ADD and USCLIST are insignificantly correlated with GIL in the predicted direction. Although

21		-
20		1 -0.025
61		$\begin{array}{c} 1 \\ 0.002 \\ 0.404^{***} \end{array}$
18		$\begin{array}{c} 1\\ 0.047\\ -0.001\\ 0.147^{***}\end{array}$
17	-	-0.134*** -0.075* -0.021 -0.117***
16	-0.003	0.018 -0.108** 0.016 -0.244***
15	-0.008	-0.064 -0.076* 0.005 -0.146***
14		-0.041 -0.1** -0.008 -0.022
13	1 0.064 0.092** 0.058	$\begin{array}{c} -0.044 \\ 0.04 \\ -0.05 \\ 0.095^{**} \end{array}$
12	1 0.005 0.135*** -0.24***	0.048 0.273*** -0.051 0.389***
11	1 0.146*** 0.146*** 0.139*** 0.065* 0.018	-0.015 -0.098** -0.044 0.073*
10	1 0.046 -0.021 -0.05 0.038 0.171 -0.05	-0.078* -0.005 0.048 -0.181***
9	1 0.067 0.058 0.05 0.097** -0.092** -0.001	$\begin{array}{c} 0.082^{*} \\ 0.02 \\ -0.051 \\ 0.128^{***} \end{array}$
8	1 -0.263*** -0.147*** 0.039 0.006 0.044 0.001 -0.044 0.001 -0.035	-0.021 -0.017 0.005 0.18***
7	1 0.057 -0.037 -0.046 -0.046 -0.144*** -0.03 0.019 0.192***	-0.043 0.096^{**} -0.008 -0.024
6	$\begin{array}{c} 1\\ 0.484\\ 0.119 \\ 0.015\\ 0.015\\ 0.015\\ 0.039 \\ -0.089 \\ -0.039 \\ -0.039 \\ -0.039 \\ -0.039 \\ -0.033 \\ -0.033 \end{array}$	$\begin{array}{c} 0.011\\ 0.027\\ 0.052\\ -0.079^{*} \end{array}$
5	$\begin{array}{c} 1\\ 0.028\\ 0.028\\ 0.034\\ 0.155^{\ast\ast\ast}\\ 0.155^{\ast\ast\ast}\\ 0.155^{\ast\ast\ast}\\ 0.155^{\ast\ast\ast}\\ 0.155^{\ast\ast\ast}\\ 0.155^{\ast\ast\ast}\\ 0.038^{\ast\ast}\\ 0.047\end{array}$	$\begin{array}{c} 0.067 \\ -0.026 \\ -0.107^{**} \\ -0.114^{***} \end{array}$
4	$\begin{array}{c} 1\\ -0.019\\ -0.0132^{***}\\ -0.053\\ 0.053\\ 0.053\\ 0.039\\ 0.039\\ 0.039\\ 0.039\\ 0.039\\ 0.039\\ 0.039\\ 0.011^{**}\\ 0.004\\ 0.011\end{array}$	0.352*** 0.012 -0.015 0.074*
3	$\begin{array}{c} 1\\ 0.216^{***}\\ -0.113^{***}\\ -0.068^{***}\\ 0.088^{***}\\ 0.0339\\ 0.035\\ 0.035\\ 0.035\\ 0.015\\ 0.003\\ 0.0015\\ 0.0064\\ 0.0064\end{array}$	0.125*** 0.006 0.002 0.058
2	$\begin{array}{c} 1\\ 0.095^{\ast\ast\ast}\\ 0.012\\ 0.012\\ 0.042\\ -0.042\\ -0.013\\ -0.184^{\ast\ast\ast}\\ -0.213\\ -0.061\\ 0.076^{\ast}\\ 0.076^{\ast}\\ 0.076^{\ast}\\ 0.063\\ 0.063\\ 0.063\\ \end{array}$	0.019 0 0.034 0.178***
1	$\begin{array}{c} 1 \\ 0.181^{+-} \\ 0.186^{+-} \\ 0.083^{+} \\ 0.083^{+} \\ 0.083^{+} \\ 0.019^{+-} \\ 0.019^{+-} \\ 0.012^{+-} \\ 0.028^{+-} \\ 0.028^{+-} \\ 0.038^{+-} \\ 0.038^{+-} \\ 0.038^{+-} \\ 0.038^{+-} \\ 0.059^{+-} \end{array}$	$\begin{array}{c} 0.054 \\ 0.061 \\ 0.024 \\ -0.023 \end{array}$
/ariable	GIL GIL CGU CGU ATURNOVER ATURNOVER ADOCF PEBRATIO DEBRATIO DEBRATIO BEATWIN A BACTWIN A BACTWIN A BACCW E SELOW	8 ADD 9 USCLIST 0 YEND 1 SIZE

Table 4. Pearson Correlations

Notes: This table presents the Pearson correlations for the variables used in the multivariate tobit regression examining the determinants of goodwill impairment losses.

The variable definitions are reported in Tables 2 and 3.

**Denotes significance at <.05 level for two-tailed test. *Denotes significance at <.10 level for two-tailed test.

***Denotes significance at <.01 level for two-tailed test.

bivariate correlations exist, the tobit analysis offers advantages over bivariate analysis on the ground of its ability to control for the effects and interrelationships between other independent variables. Therefore, the results of the multivariate regression are given greater consideration in this study.

6.3. Multivariate Results

Table 5 presents the results of the multivariate tobit regression with an adjusted R^2 of 13.2 per cent. In terms of proxies for actual (economic) impairment, as predicted in *H1*, *B/M* (*Z* = 1.960) is positive and significant. $\triangle OCF$ (*Z* = -1.762) and *ROA* (*Z* = -1.819) are significantly negative while *GWA* (*Z* = 1.358), albeit of the correct sign, is insignificant. *CGU* (*Z* = 1.455) is positive and insignificant. Finally, contrary

Variable	Prediction	Coefficient	Z-statistic	p-value	VIF
Intercept		-0.1689	-6.279	<.001	
B/M	+	0.0203	1.960	.048	1.339
ĠWA	+	0.0216	1.358	.174	1.154
CGU	?	0.0131	1.455	.145	1.259
$\Delta TURNOVER$	_	0.0062	0.940	.347	1.232
ΔOCF	_	-0.0589	-1.762	.078	1.647
ROA	_	-0.0533	-1.819	.068	1.692
DEBTRATIO	?	-0.0121	-0.711	.477	1.380
BATH	_	-0.0938	-3.381	<.001	1.872
SMOOTH	+	0.1019	2.136	.0327	1.189
ΔCEO	+	0.0105	1.732	.079	1.097
BINDEP	+	0.0460	2.130	.032	1.575
SEPCHAIR	+	0.0060	0.479	.623	1.086
BACTIVITY	+	0.0033	3.845	<.001	1.201
BLOCK	+	0.0005	2.675	.0075	1.181
EXEOWN	+	0.0596	2.047	.046	1.261
NONEXEOWN	+	0.1102	2.393	.016	1.117
ADD	+	0.0171	2.276	.022	1.259
USCLIST	+	0.0201	1.964	.049	1.434
YEND	?	0.0097	1.654	.120	1.038
SIZE	?	0.0025	1.183	.237	2.739
McFadden's Adjusted R^2	0.132				

Table 5. Multivariate Tobit Regression: The Determinants of GoodwillImpairment Losses

Notes: This table presents the results of the tobit regression examining the determinants of goodwill impairment losses. The model uses a sample of 528 firm-year observations (109 write-off observations and 419 non-write-off observations). The variable definitions are reported in Tables 2 and 3. In this table, the bold values indicate significance.

to the expectations, $\Delta TURNOVER$ (Z = 0.940) is positive and insignificant. In terms of reporting incentive, as predicted in H2b and H2c, BATH (Z = -3.381) is negative and significant and both SMOOTH (Z = 2.136) and ΔCEO (Z = 1.732) are significantly positive.

DEBTRATIO (Z = -0.711), is negative and insignificant, suggesting that debt is less likely to provide U.K. managers with incentives to manipulate the amount of goodwill impairment losses to avoid the costly violations of debt covenants compared to their U.S. and Canadian counterparts. This may be because there is less information asymmetry between U.K. managers and (private) lenders which creates fewer incentives to manage earnings.²⁰ In terms of corporate governance variables, *BINDEP* (Z = 2.130), *BACTIVITY* (Z = 3.845), *BLOCK* (Z = 2.675), *EXEOWN* (Z = 2.047), and *NONEXEOWN* (Z = 2.393) are significantly positive, as predicted in *H3. SEPCHAIR* (Z = 0.479), while of the correct sign, is insignificant. Finally, in terms of control variables, as predicted, *ADD* (Z = 2.276) and *USCLIST* (Z = 1.964) are positive and significant, while, *YEND* (Z = 1.654) and *SIZE* (Z = 1.183) are positive and insignificant.

After controlling for economic factors, the overall results suggest that managers are exercising discretion in the reporting of goodwill impairment losses following the adoption of IFRS 3, as indicated by the significant coefficients for *BATH*, *SMOOTH*, and ΔCEO . However, the results also reveal that goodwill impairments are strongly associated with effective governance mechanisms. This suggests that managers are more likely to be exercising their accounting discretion to convey their private information and expectations about the underlying performance of the firm than to be acting opportunistically. The results collectively suggest that, in recording goodwill impairments following the adoption of IFRS 3, managers are more likely responding to changes in economic circumstances and real declines in the value of the firm.

6.4. Factorization of Corporate Governance Variables

Without a clear theory about the complex, multi-dimensional nature of corporate governance and in order to develop a parsimonious representation for this construct and draw stronger conclusions regarding its role in constraining managerial opportunism over write-offs (Larcker et al., 2007), principal component analysis is also used in this study as a sensitivity analysis to group the individual corporate governance variables into more reliable and valid factors that capture different dimensions of corporate governance quality and yield more stable estimates. *SEPCHAIR* variable is omitted from this analysis as it was not significant in explaining managers' goodwill reporting choices when used individually. Factors with an eigenvalue greater than one are retained (Kaiser, 1960). Varimax orthogonal (uncorrelated) rotation is used to minimize the number of variables that have high loadings on each factor.

Table 6A, B present the results of this analysis. Three factors (principal components) that jointly explain 71.023 per cent of the total variance in the individual corporate governance proxies are retained.²¹

 Table 6. Corporate Governance Variables Based on Principal Component

 Analysis

Variable	Principal component 1	Principal component 2	Principal component 3
BINDEP	0.419	-0.617	-0.292
BACTIVITY	0.749	0.061	0.075
BLOCK	0.026	-0.002	0.968
EXEOWN	-0.751	0.082	0.062
NONEXEOWN	0.141	0.890	-0.116

(B) Description

(A) Factor loading

	CG-PC1 monitoring versus managerial ownership	CG-PC2 non-executive directors' ownership versus board independence	CG-PC3 blockholders
Eigenvalue	1.460	1.091	1.000
Variance explained	29.204	21.811	20.008
Variance explained (cumulative)	29.204	51.015	71.023

Notes: Components have been Varimax rotated. Rotation converged in five iterations.

This table presents the details about the principal component analysis used to factorize governance variables into a smaller number of principal components (factors). (A) Reports the factor loadings for the retained three principal components. Figures in bold print indicate factor loadings in excess of 0.4 in absolute terms.

(B) Reports the names assigned to each principal component based on the governance variables which exhibit factor loadings in excess of 0.40 in absolute value. The bottom three rows of (B) present the eigenvalues assigned for each principal component and the percentage of total variance in the governance variables explained by each principal component separately and cumulatively. The principal component analysis is based on the 528 firm-year observations used in the primary analysis of the study.

Following Larcker et al. (2007), variables are retained if they have a minimum factor loading of 0.40 in absolute value. Similar to Larcker et al. (2007), each factor is assigned a name based on the characteristics of the governance variables which exhibit factor loadings in excess of 0.40 in absolute value. The first factor is positively correlated with BINDEP and BACTIVITY, and negatively correlated with EXEOWN. This factor seems to account for the monitoring power of the board of directors as reflected by board independence and activity versus managerial ownership. The second factor is positively correlated with NON-EXEOWN and negatively correlated with BINDEP. This factor seems to reflect concern for outside directors' interest versus board independence. Finally, the third factor is positively correlated with BLOCK and hence seems to account for the monitoring power of institutional (large) shareholders.²² The primary tobit analysis is repeated after replacing the individual corporate governance variables with the three corporate governance factors. Consistent with the hypothesis examined in this study, a positive association is expected between reported goodwill impairment losses and each of these factors. Table 7 presents the results of this analysis. The first (Z = 2.596) and third (Z = 1.783) governance factors are positive and significant at the 1 and 10 per cent levels, respectively. The second governance factor (Z = 1.140), while of the correct sign, is insignificant. The inferences on all the other variables remain unaffected and similar to those reported in Table 5. These results provide corroborating evidence regarding the role corporate governance plays in mitigating managerial opportunism with respect to goodwill impairment losses and reinforce the conclusions reached in the primary analysis of the study.

7. Additional Analyses

To provide additional assurance that the perceived reporting incentives are less likely to be opportunistic. The primary tobit analysis was repeated including the interactions between *BATH*, *SMOOTH*, and ΔCEO with the first principal component (*CG-PC1*).²³ This was under the hypothesis that these reporting incentives are less likely to be opportunistic if they occur in conjunction with effective governance mechanisms. Earnings management will benefit shareholders if managers use their accounting discretion to signal private information about future performance (Healy and Wahlen, 1999). In such cases, shareholders will not expect boards to constrain earnings management.

Variable	Prediction	Coefficient	Z-statistic	p-value	VIF
Intercept		-0.0979	-5.324	<.001	
B/M	+	0.0217	1.995	.046	1.332
ĠWA	+	0.0182	1.105	.269	1.153
CGU	?	0.0145	1.584	.113	1.244
$\Delta TURNOVER$	_	0.0024	0.366	.714	1.208
ΔOCF	_	-0.0420	-1.642	.084	1.622
ROA	_	-0.0634	-2.383	.017	1.676
DEBTRATIO	?	-0.0202	-1.170	.242	1.349
BATH	_	-0.1034	-3.644	<.001	1.829
SMOOTH	+	0.1232	2.910	.004	1.169
ΔCEO	+	0.0133	2.218	.023	1.085
CG-PC1	+	0.0078	2.596	.009	1.259
CG-PC2	+	0.0033	1.140	.254	1.215
CG-PC3	+	0.0050	1.783	.074	1.207
ADD	+	0.0151	1.990	.047	1.245
USCLIST	+	0.0200	1.859	.063	1.393
YEND	?	0.0086	1.550	.121	1.029
SIZE	?	0.0027	0.946	.344	2.632
McFadden's Adjusted R^2	0.153				

 Table 7. Tobit Regression Using Corporate Governance Factors

Notes: This table presents the results of the tobit regression examining the determinants of goodwill impairment losses. This model is similar to the model shown in Table 5 except that corporate governance variables (*BINDEP*, *BACTIVITY*, *BLOCK*, *EXEOWN*, *NONEXE-OWN*) are replaced with corporate governance factors (*CG-PC1*, *CG-PC2*, *CG-PC3*) based on principal component analysis and *SEPCHAIR* is omitted. The model uses a sample of 528 firm-year observations over the period 2005–2006 (109 write-off observations and 419 non-write-off observations).

Variable definitions: CG-PC1: Corporate governance principal component 1 representing board independence and activity versus managerial ownership.

CG-PC2: Corporate governance principal component 2 representing non-executive directors' ownership versus board independence.

CG-PC3: Corporate governance principal component 3 representing blockholders.

The other variables are defined in Tables 2 and 3.

A negative sign is predicted on $BATH^*CG$ -PC1 and a positive sign on both $SMOOTH^*CG$ -PC1 and ΔCEO^*CG -PC1. Consistent with this hypothesis, results (untabulated) reveal that $BATH^*BINDEP$ (Z = -3.867), $SMOOTH^*BINDEP$ (Z = 2.763) and $\Delta CEO^*BINDEP$ (Z = 2.138) are all significant in the predicted direction. The inferences on all the other variables remain unaffected, providing additional evidence that the perceived reporting incentives more likely reflect the provision of managers' private information, as opposed to their acting opportunistically.

As discussed earlier, the use of OLS when the dependent variable is censored is inappropriate and produces biased (underestimated in particular) as well as inconsistent estimates (Greene, 1997; Gujarati,

2003). However, Maddala (1991, p. 804) notes that the tobit model should not be used if the observations on the dependent variable are missing because of individual choices rather than pure censoring. From this perspective, it is possible that some or even all of the non-write-off observations have true values of zero, reflecting that there has been no change in the economic value of their goodwill (no increase or decrease), and suggesting that the distribution may not be censored. In this alternative scenario, the use of OLS may be appropriate. Consequently, an OLS regression was also run. The untabulated results (adjusted $R^2 = 18.6$ per cent) are quite similar to those reported under the primary tobit analysis. However, B/M, $\triangle CEO$, BLOCK, and NON-EXEOWN, while all of the correct sign, are no longer significant; and GWA, while insignificant in the primary analysis, become significant at the 10 per cent level. The inferences on all the other variables remain unaffected. Comparison of the OLS and the tobit coefficients, reveals that OLS coefficients are smaller in magnitude than their tobit equivalents, consistent with the view that the use of OLS when the dependent variable is censored produces biased (underestimated in particular) and inconsistent estimates.

To examine whether the results of the study are driven by industry effects, the primary tobit analysis was repeated after incorporating eight industry dichotomous variables. The industry membership is defined by the Industry Classification Benchmark (ICB) code, as given by the London Stock Exchange (LSE). The industry of the firm is expected to provide some common effect with respect to the goodwill impairment, as the impairment charges are closely related to competition, deterioration, or other economic factors of the industry (Bens, 2006; Zang, 2008). However, the untabulated results reveal that none of the industries is significant in explaining the goodwill impairment losses, suggesting that an industry effect is unlikely to have driven the results documented in the primary tobit analysis. The inferences on all the other variables remain unaffected. This result is also consistent with the write-off and non-write-off observations having a similar industry composition with three industries, Consumer Services (ICB 5000), Industrials (ICB 2000), and Consumer Goods (ICB 3000), having the highest presentation (approximately 74 per cent) within each grouping (untabulated).

Unlike prior studies (e.g., Beatty and Weber, 2006; Lapointe-Antunes et al., 2008), the current study does not examine the role of managers' bonuses in their decisions to report goodwill impairment

losses. Conyon et al. (2006) document that base salaries are a more important component of CEO pay in the United Kingdom compared to the United States. They find that U.S. CEO incentives in 2003 were about 8.8 times U.K. CEO incentives, suggesting that pay-related governance problems are more severe in the United States. However, the analysis was repeated after adding *BONUS* as measured by the value of bonus compensation for the CEO at the end of *t* divided by the CEO's base salary at the end of *t*. Untabulated results reveal that *BONUS* (Z = 1.503) is positive and insignificant suggesting that U.K. managers are less likely to exploit their accounting discretion to manage earnings for bonus purposes.

Finally, in addition to the pooled regression analysis, yearly analysis was performed to examine whether the results are time-sensitive or influenced by transition effects from U.K. GAAP to IFRS. The untabulated results yield inferences that are qualitatively similar to those reported under the primary tobit analysis (Table 5).

8. Summary and Conclusions

Using a sample of 528 firm-year observations, drawn from the top 500 U.K. listed firms for the years 2005 and 2006, this study examines managers' use of discretion in determining goodwill impairment losses following the mandatory adoption of IFRS 3, and whether this discretion reflects opportunistic reporting by managers or the provision of their private information. Despite the standard setters' contention that the impairment-only approach will improve the accounting treatment for goodwill and provide users with more useful and value-relevant information regarding the underlying economic value of goodwill, this approach has been largely criticized on the grounds of the managerial discretion inherent in the process of testing goodwill for impairment. Therefore, *ex-ante*, it is unclear how the impairment-only approach has affected the characteristics of reported goodwill impairment losses.

After controlling for economic factors, empirical results reveal that managers are exercising discretion in the reporting of goodwill impairment losses following the adoption of IFRS 3. The results also reveal that goodwill impairments are strongly associated with effective governance mechanisms, suggesting that managers are more likely to be using the discretion afforded by IFRS 3 efficiently to convey their private information and expectations about the underlying performance of the firm rather than acting opportunistically in the write-off year.

Effective governance mechanisms are likely to restrict managers' ability to report goodwill impairments that differ from predicted economic losses, resulting in the recognition of more timely impairments that better reflect the firm's underlying economics. Contrary to criticisms that surrounded the application of the impairment approach, the overall results suggest that IFRS No. 3 has improved the quality of reported goodwill impairment losses, and hence provide support to the IASB's contention that IFRS 3 allows firms to reflect their underlying economic attributes. These inferences are robust to various modeling specifications and variable definitions. They are also robust to the principal component aggregation of the corporate governance variables.

One of the criticisms of write-offs reported under the former triggerbased standards (e.g., FRS 10 and FRS 11) is that an absence of a specific impairment trigger gave firms too much discretion in timing the write-offs which could be used by managers opportunistically to meet certain reporting objectives (Henning et al., 2004; Hayn and Hughes, 2006). It is possible that the requirement to test goodwill for impairment annually (irrespective of whether there is any indication that it may be impaired) at the cash-generating unit level has removed many of the "cushions" protecting goodwill from impairment (IASB, 2004a). It could also have eradicated the discretion available to managers in deciding when to test it for impairment, resulting in more timely recognitions of existing goodwill impairments that better reflect the underlying performance of the firm.

The results of this study should be of interest to standard setters and policy makers as they suggest that managers are likely to utilize the discretion permitted in principles-based standards. They also highlight the importance of effective governance mechanisms in constraining managerial opportunism associated with such standards. Future research may examine the value relevance or information content of IFRS 3 goodwill impairments. Such studies offer an opportunity to explore additional questions related to the relevance of the goodwill impairments and hence may be viewed as complementary to this study.

Notes

1. For example, in explaining how IFRS 3 improves financial reporting, the IASB (2004a, BC 140, 142) argued that "straight-line amortization of goodwill over an arbitrary period fails to provide useful information. The Board noted that both anecdotal

and research evidence supports this view ... The Board reaffirmed the view it reached in developing ED 3 that if a rigorous and operational impairment test could be devised, more useful information would be provided to users of an entity's financial statements under an approach in which goodwill is not amortized, but tested for impairment annually or more frequently if events or changes in circumstances indicate that the goodwill might be impaired."

2. FRS 10 allowed goodwill to have a useful economic life of greater than 20 years, or even indefinite, but only when it is expected to be capable of continued measurement. Where goodwill is regarded as having an indefinite useful economic life, it should not be amortized. If goodwill is not amortized, or amortized over a period of more than 20 years, then an impairment review must be performed each year to ensure that the carrying value of the goodwill does not exceed its recoverable amount in accordance with FRS 11 *Impairment of Fixed Assets and Goodwill* (ASB, 1998). However, the way that U.K. firms applied the requirements of FRS 10 and FRS 11 was regarded as "slightly surprising, given their long-standing hostility to amortizing goodwill: most of them chose the amortization route in order to avoid the complexities of the full-blown impairment testing regime" (Paterson, 2002, p. 102). Andrews (2006) also reports that the majority of large U.K. firms in the 2004 financial year have selected 20 years as the finite useful economic life for goodwill and have amortized the asset over its finite life.

3. In deciding not to converge with SFAS 142 on the level of the goodwill impairment test, the Board noted that several North American round-table participants expressed a high level of dissatisfaction at being prevented by SFAS 142 from recognizing goodwill impairments that they knew existed at levels lower than reporting units (as defined by SFAS 142), but which disappeared once the lower level units were aggregated with other units containing sufficient cushions to offset the impairment loss (IASB, 2004b, BC 149).

4. The goodwill impairment test required by IFRS 3 is a one-stage process, as opposed to the two-stage test required by the U.S. SFAS 142. If the carrying amount of a cash-generating unit that contains goodwill exceeds its fair value, firms are required to report an impairment loss under IAS 36. However, under SFAS 142, U.S. firms that fail the first step can still avoid recording an impairment loss if the implied fair value of goodwill exceeds its carrying value (FASB, 2001).

5. For example, managers may have incentives to act strategically in the transition period by increasing the amount of write-offs that are treated as merely an accounting change or charged to retained earnings thereby decreasing the probability and amount of future impairments that would, if recorded, be included in income from continuing operations (Beatty and Weber, 2006).

6. Ramanna (2008) argues that these findings can alternatively be explained by arguments that impairments are utilized as a "big bath" strategy, or by management's incompetence to avoid losses despite SFAS 142's discretion potential. In both cases, the impairments are informative to markets, but not because the standard has provided a framework for managers to reliably report private information.

7. The implicit assumption in this study is that causality runs from corporate governance practices to reported impairment losses. However, corporate governance practices may also be driven by the poor performance that necessitates goodwill impairment losses. Endogeneity problems in corporate governance research may be large but not "fatal" requiring careful interpretation of the results (Denis, 2001, p. 198).

8. The one-stage impairment test of IFRS 3 suggests that managers cannot evaluate whether impairment occurs without understanding the magnitude of the impairment. The alternative two-stage specification is more appropriate in a U.S. context, with the

first stage capturing the decision to report an impairment using a logistics regression, and the second capturing the magnitude of the impairment using a tobit regression.

9. A tobit regression is used rather than an OLS. The tobit is a censored regression model where observations on the dependent variable are unobservable below some threshold (Maddala, 1983, 1991). Firms that experience an increase in the economic value of goodwill are not allowed to record the increase, causing the distribution of the dependent variable to be censored at zero. OLS produces biased and inconsistent estimates when the dependent variable is censored (Greene, 1997; Gujarati, 2003).

10. Due to the difficulty of separating write-offs into discretionary and non-discretionary components, the majority of write-off studies (e.g., Francis et al., 1996; Riedl, 2004; Beatty and Weber, 2006; Zang, 2008) use the reported write-off amount as the dependent variable, assuming all write-offs to be unanticipated.

11. This may be a noisy measure as firms that perform poorly at the firm level may be able to avoid impairment if the cash-generating unit performs particularly well. Alternatively, a firm may perform well but still have cash-generating units that impair goodwill because of poor performance.

12. The study deflates all firm-specific financial variables by the same deflator (lagged total assets) to reduce potential heteroscedasticity problems (e.g., Riedl, 2004; Lapointe-Antunes et al., 2008; Zang, 2008).

13. *CGU* is measured as a dichotomous variable rather than a continuous one as many firms do not disclose the number of cash-generating units, simply stating that goodwill is allocated to "multiple" units. This measurement is consistent with Beatty and Weber (2006).

14. Consistent with prior research (Beatty and Weber, 2006; Lapointe-Antunes et al., 2008), this study considers only changes in CEO, as CEOs are more likely to have incentives to accelerate or delay goodwill impairments depending on whether they made the acquisition decision or not. The study does not also differentiate between forced and voluntary CEO changes as in most cases companies do not disclose the reason for the change.

15. Shleifer and Vishny (1997) suggest that large shareholders with substantial ownership stakes, such as 10 or 20 per cent, are more likely to have incentives to collect information and monitor the management. The measurement of *BLOCK* is consistent with this view. An alternative definition of *BLOCK* is examined: *BLOCK1* equals the cumulative percentage of outstanding common shares held by blockholders holding at least 3 per cent of outstanding shares and who are not part of the board of directors. Untabulated results reveal that *BLOCK1* (Z = 1.152), although still positive, is no longer significant. The inferences on all the other variables remain unaffected and similar to those reported in the primary tobit analysis. This result provides corroborating evidence consistent with the view of Shleifer and Vishny (1997) adopted in this study.

16. Another explanation derived from the transaction cost theory suggests that crosslisted firms would incur fewer extra compliance costs than non-cross-listed firms because they have been required to carry out goodwill impairment tests following the requirements of SFAS 142.

17. Similar to prior U.S. and Canadian research (e.g., Beatty and Weber, 2006; Lapointe-Antunes et al., 2008; Zang, 2008), the analysis in this study is limited to the post-adoption period only. As such, the purpose is not to measure the impact of standard change but the effects of the mandatory IFRS 3 application on reported goodwill impairment losses. However, the limited number of years studied is a limitation of the current research. Given more years of financial statement data, it may be possible to examine the long-term effects of IFRS 3 on goodwill accounting and determine whether the conclusions of this study hold over time. Furthermore, sample firms come from listed companies in a single country, and further investigations using listed firms in other countries are warranted.

18. This is expected as the study excludes firms listed on the Alternative Investment Market, which is primarily a small-cap market, and firms with no positive goodwill balances which are expected to be smaller than firms active in mergers and acquisitions.

19. The highest variance inflation factor is less than 3, providing further evidence that multicollinearity is not a problem in this study.

20. In the United Kingdom, Ball et al. (2000) provide evidence that corporate debt is predominately private which reduces the information asymmetry between managers and (private) lenders. When information asymmetry is low, lenders are more likely to have a closer relationship with firms (borrowers) providing the former with the necessary information to monitor the latter on their actions and hence leaving less room for earnings management around debt contracts.

21. The Bartlett's Test of Spherity is significant (p < .001) (untabulated) indicating that the factor model is appropriate.

22. *BLOCK* loads heavily and individually to the third factor suggesting that large shareholders form a unique governance mechanism that acts independently of other mechanisms.

23. Interactions are performed only with CG-PCI to avoid having a complex model. The first principal component explains 29.2 per cent of total variation, which exceeds the explanatory power of the second and third components (21.8 and 20.0 per cent, respectively).

References

- Accounting Standards Board, Financial Reporting Standard 10, *Goodwill and Intangible Assets*, (London: ASB, 1997).
- Accounting Standards Board, Financial Reporting Standard 11, Impairment of Fixed Assets and Goodwill, (London: ASB, 1998).
- Accounting Standards Committee, Statement of Standard Accounting Practice 22, *Accounting for Goodwill*, (London: ASC, 1984, 1989).
- Alciatore, M., C.C. Dee, P. Easton and N. Spear, "Asset Write-Downs: A Decade of Research," *Journal of Accounting Literature* 17 (1998), pp. 1–39.
- Andrews, R., Impairment of Assets: Measurement Without Disclosure, The Chartered Association of Certified Accountants Research Report 92, (2006).
- Ball, R., S.P. Kothari and A. Robin, "The Effect of International Institutional Factors on Properties of Accounting Earnings," *Journal of Accounting and Economics* 29 (2000), pp. 1–51.
- Beasley, M.S., "An Empirical Analysis of the Relation between the Board of Director Composition and Financial Statement Fraud," *The Accounting Review* 71 (1996), pp. 443–465.
- Beatty, A. and J. Weber, "Accounting Discretion in Fair Value Estimates: An Examination of SFAS 142 Goodwill Impairments," *Journal of Accounting Research* 44 (2006), pp. 257–288.
- Bens, D.A., "Discussion of Accounting Discretion in Fair Value Estimates: An Examination of SFAS 142 Goodwill Impairments," *Journal of Accounting Research* 44 (2006), 289–296.
- Bushee, B.J., "The Influence of Institutional Investors on Myopic R&D Investment Behavior," *The Accounting Review* 73 (1998), pp. 305–333.

- Chen, C., M. Kohlbeck and T. Warfield, "Goodwill Valuation Effects of the Initial Adoption of SFAS 142," Available at SSRN: http://ssrn.com/abstract=534484, (1 April 2004).
- Christie, A.A. and J.L. Zimmerman, "Efficient and Opportunistic Choices of Accounting Procedures: Corporate Control Contests," *The Accounting Review* 69 (1994), pp. 539–566.
- Chtourou, S.M., J. Bedard and L. Courteau, "Corporate Governance and Earnings Management," Available at SSRN: http://ssrn.com/abstract=275053, (April 21, 2001).
- Conyon, M.J., J.E. Core and W.R. Guay, "How High is US CEO Pay? A Comparison with UK CEO Pay," *Industrial Organization Abstracts: Firm Structure, Purpose, Organization & Contracting*, Working Paper Series, vol. 7, no. 11 (July 11, 2006).
- Dechow, P.M., R.G. Sloan and A.P. Sweeney, "Causes and Consequences of Earnings Manipulation: An Analysis of Firms Subject to Enforcement Actions by the SEC," *Contemporary Accounting Research* 13 (1996), pp. 1–36.
- Denis, D.K., "Twenty-Five Years of Corporate Governance Research...and Counting," *Review of Financial Economics* 10 (2001), pp. 191–212.
- Ebrahim, A., "Earnings Management and Board Activity: An Additional Evidence," *Review of Accounting and Finance* 6 (2007), pp. 42–58.
- Elliott, J.A. and W.H. Shaw, "Write-Offs as Accounting Procedures to Manage Perceptions," *Journal of Accounting Research*, 26(Suppl.) (1988), pp. 91–119.
- Fama, E.F. and M.C. Jensen, "Separation of Ownership and Control," Journal of Law and Economics 26 (1983), pp. 310–325.
- Fields, T.D., T.Z. Lys and L. Vincent, "Empirical Research on Accounting Choice," *Journal of Accounting and Economics* 31 (2001), pp. 255–307.
- Financial Accounting Standards Board, Statement of Financial Accounting Standards No. 142, *Goodwill and Other Intangible Assets*, (Norwalk, CT: FASB, 2001).
- Financial Reporting Council, The Combined Code on Corporate Governance (London, UK: Financial Reporting Council, 2003).
- Francis, J., J. Hanna and L. Vincent, "Causes and Effects of Discretionary Asset Write-Offs," *Journal of Accounting Research* 34 (Suppl.) (1996), pp. 117–134.
- Godfrey, J.M. and P. Koh, "Goodwill Impairment as a Reflection of Investment Opportunities," *Accounting and Finance* 49 (2009), pp. 117–140.
- Greene, W.H., *Econometric Analysis*, 3rd edition (Upper Saddle River, NJ: Prentice Hall, 1997).
- Gujarati, D.N., Basic Econometrics, 4th edition (London: McGraw-Hill, 2003).
- Hayn, C. and P. Hughes, "Leading Indicators of Goodwill Impairment," Journal of Accounting, Auditing and Finance 21 (2006), pp. 223-265.
- Healy, P.M. and J.M. Wahlen, "A Review of the Earnings Management Literature and Its Implications for Standard Setting," *Accounting Horizons* 13 (1999), pp. 365–383.
- Henning, S.L., W.H. Shaw and T. Stock, "The Amount and Timing of Goodwill Write-Offs and Revaluations: Evidence from US and UK Firms," *Review of Quantitative Finance and Accounting* 23 (2004), 99–121.

Higson, C., "Goodwill," British Accounting Review, 30 (1998), pp. 141-158.

- Holthausen, R.W., "Accounting Method Choice: Opportunistic Behavior, Efficient Contracting, and Information Perspectives," *Journal of Accounting and Economics* 12 (1990), pp. 207–218.
- Huijgen, C. and M. Lubberink, "Earnings Conservatism, Litigation and Contracting: The Case of Cross-Listed Firms," *Journal of Business Finance and Accounting* 32(7–8) (2005), pp. 1275–1309.

- Hussey, R. and A. Ong, "Can We Put a Value on a Name: The Problem of Accounting for Goodwill and Brands," *Credit Control* 21 (2000), pp. 32–38.
- International Accounting Standards Board, International Financial Reporting Standard 3, *Business Combinations*, (London: IASB, 2004a).
- International Accounting Standards Board, International Accounting Standard 36 (revised), *Impairment of Asset*, (London: IASB, 2004b).
- Jarva, H., "Do Firms Manage Fair Value Estimates? An Examination of SFAS 142 Goodwill Impairments," *Journal of Business Finance and Accounting* 36 (2009), pp. 1059–1086.
- Jensen, M.C., "Eclipse of the Public Corporation," *Harvard Business Review* 67(September–October) (1989), pp. 61–74.
- Jensen, M.C., "The Modern Industrial Revolution, Exit, and the Failure of Internal Control Systems," *The Journal of Finance* 48 (1993), pp. 831–880.
- Jensen, M.C. and W.H. Meckling, "Theory of Firm: Managerial Behavior, Agency Costs and Ownership Structure," *Journal of Financial Economics* 3 (1976), pp. 305– 360.
- Kaiser, H., "The Application of Electronic Computers to Factor Analysis," *Educational* and *Psychological Measurement* 20 (1960), pp. 141–151.
- Karamanou, I. and N. Vafeas, "The Association Between Corporate Boards, Audit Committees, and Management Earnings Forecasts: An Empirical Analysis," *Journal* of Accounting Research 43 (2005), pp. 453–486.
- Kirschenheiter, M. and N.D. Melumad, "Can "Big Bath" and Earnings Smoothing Co-Exist as Equilibrium Financial Reporting Strategies?" *Journal of Accounting Research* 40 (2002), pp. 761–796.
- Klein, A., "Audit Committee, Board of Director Characteristics, and Earnings Management," *Journal of Accounting and Economics* 33 (2002), pp. 375–400.
- Koh, P., "On the Association Between Institutional Ownership and Aggressive Corporate Earnings Management in Australia," *The British Accounting Review* 35 (2003), pp. 105–128.
- Kvaal, E., Topics in Accounting for Impairment of Fixed Assets, PhD thesis, (Norway: BI Norwegian School of Management, 2005).
- Lapointe-Antunes, P., D. Cormier and M. Magnan, "Equity Recognition of Mandatory Accounting Changes: The Case of Transitional Goodwill Impairment Losses," *Canadian Journal of Administrative Sciences* 25 (2008), pp. 37–54.
- Lapointe-Antunes, P., D. Cormier and M. Magnan, "Value Relevance and Timeliness of Transitional Goodwill-Impairment Losses: Evidence from Canada," *International Journal of Accounting* 44 (2009), pp. 56–78.
- Larcker, D.F., S.A. Richardson and I. Tuna, "Corporate Governance, Accounting Outcomes, and Organizational Performance," *The Accounting Review* 82 (2007), pp. 963– 1008.
- Li, Z., P.K. Shroff, R. Venkataraman and I. Zhang, "Causes and Consequences of Goodwill Impairment Losses," AAA 2005 FARS Meeting Paper. Available at SSRN: http://ssrn.com/abstract = 590908, (May 1, 2010).
- Maddala, G.S., *Limited-Dependent and Qualitative Variables in Econometrics* (New York: Cambridge University Press, 1983).
- Maddala, G.S., "A Perspective on the Use of Limited-Dependent and Qualitative Variables Models in Accounting Research," *The Accounting Review* 66 (1991), pp. 788– 807.
- Massoud, M.F. and C.A. Raiborn, "Accounting for Goodwill: Are we Better Off?" *Review of Business* 24 (2003), pp. 26–32.

- Masters-Stout, B., M.L. Costigan and L.M. Lovata, "Goodwill Impairments and Chief Executive Officer Tenure," *Critical Perspectives on Accounting* 19 (2008), pp. 1370– 1383.
- Moore, M.L., "Management Changes and Discretionary Accounting Decisions," Journal of Accounting Research 11 (1973), pp. 100–107.
- Morck, R., A. Shleifer and R.W. Vishny, "Managerial Ownership and Market Valuation: An Empirical Analysis," *Journal of Financial Economics* 20 (1988), pp. 292–315.
- Mulgrew, M. and J. Forker, "Independent Non-Executive Directors and Earnings Management in the UK," *The Irish Accounting Review* 13 (2006), pp. 35–64.
- Murphy, K.J. and J.L. Zimmerman, "Financial Performance Surrounding CEO Turnover," Journal of Accounting and Economics 16 (1993), pp. 273–315.
- Nobes, C., "A Political History of Goodwill in the UK: An Illustration of Cyclical Standard Setting," *ABACUS* 28 (1992), pp. 142–167.
- Paterson, R., "Straining Goodwill," Accountancy, London 129 (2002), pp. 101-103.
- Peasnell, K.V., P.F. Pope and S. Young, "Board Monitoring and Earnings Management: Do Outside Directors Influence Abnormal Accruals?" *Journal of Business Finance and Accounting* 32 (2005), pp. 1311–1346.
- Ramanna, K., "The Implications of Unverifiable Fair-Value Accounting: Evidence from the Political Economy of Goodwill Accounting," *Journal of Accounting and Economics* 45 (2008), pp. 253–281.
- Ramanna, K. and R.L. Watts, "Evidence on the Use of Unverifiable Estimates in Required Goodwill Impairment," Harvard Business School Accounting & Management Unit Working Paper No. 09–106. Available at http://ssrn.com/abstract = 1134943, (2009).
- Rees, L., S. Gill and R. Gore, "An Investigation of Asset Write-Downs and Concurrent Abnormal Accruals," *Journal of Accounting Research* 34 (Suppl.) (1996), pp. 157–169.
- Riedl, E.J., "An Examination of Long-Lived Asset Impairments," *The Accounting Review* 79 (2004), pp. 823–852.
- Schneider, C., "Four Ways to Say Goodbye to Goodwill Amortisation: Expert Tips for Tackling the Impairment Test," Available at http://www.cfo.com/article.cfm/ 2997501?f=related, (2001).
- Shleifer, A. and R.W. Vishny, "Large Shareholders and Corporate Control," *The Journal of Political Economy* 94 (1986), pp. 461–488.
- Shleifer, A. and R.W. Vishny, "A Survey of Corporate Governance," *The Journal of Finance* 52(2) (1997), pp. 737–783.
- Short, H. and K. Keasey, "Managerial Ownership and the Performance of Firms: Evidence from the UK," *Journal of Corporate Finance* 5 (1999), pp. 79–101.
- Strong, J.S. and J.R. Meyer, "Asset Writedowns: Managerial Incentives and Security Returns," *The Journal of Finance* 42 (1987), pp. 643–661.
- Vafeas, N., "Board Meeting Frequency and Firm Performance," Journal of Financial Economics 53 (1999), pp. 113–142.
- Verriest, A. and A. Gaeremynck, "What Determines Goodwill Impairment," *Review of Business and Economics* LIV (2009), pp. 106–128.
- Warfield, T.D., J.J. Wild and K.L. Wild, "Managerial Ownership, Accounting Choices, and Informativeness of Earnings," *Journal of Accounting and Economics* 20 (1995), pp. 61–91.
- Watts, R.L., "Conservatism in Accounting. Part 1: Explanations and Implications," Accounting Horizons 17 (2003), pp. 207–221.
- Watts, R.L. and J.L. Zimmerman, *Positive Accounting Theory* (Englewood Cliffs, New Jersey: Prentice-Hall, 1986).

- Watts, R.L. and J.L. Zimmerman, "Positive Accounting Theory: A Ten Year Perspective," *The Accounting Review* 65 (1990), pp. 131–156.
- Wilson, G.P., "Discussion Write-Offs: Manipulation or Impairment," Journal of Accounting Research 34(Suppl.), (1996), pp. 171–177.
- Xie, B., W.N. Davidson III and P.J. DaDalt, "Earnings Management and Corporate Governance: The Role of the Board and the Audit Committee," *Journal of Corporate Finance* 9 (2003), pp. 295–316.
- Zang, Y., "Discretionary Behaviour with Respect to the Adoption of SFAS No. 142 and the Behaviour of Security Prices," *Review of Accounting and Finance* 7 (2008), pp. 38–68.
- Zucca, L.J. and D.R. Campbell, "A Closer Look at Discretionary Writedowns of Impaired Assets," *Accounting Horizons* 6 (1992), pp. 30–41.